

Temi di Discussione

(Working Papers)

BTP futures and cash relationships: a high frequency data analysis

by Onofrio Panzarino, Francesco Potente and Alfonso Puorro





Temi di discussione

(Working papers)

BTP futures and cash relationships: a high frequency data analysis

by Onofrio Panzarino, Francesco Potente and Alfonso Puorro

Number 1083 - September 2016

The purpose of the Temi di discussione series is to promote the circulation of working papers prepared within the Bank of Italy or presented in Bank seminars by outside economists with the aim of stimulating comments and suggestions.

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

Editorial Board: Pietro Tommasino, Piergiorgio Alessandri, Valentina Aprigliano, Nicola Branzoli, Ines Buono, Lorenzo Burlon, Francesco Caprioli, Marco Casiraghi, Giuseppe Ilardi, Francesco Manaresi, Elisabetta Olivieri, Lucia Paola Maria Rizzica, Laura Sigalotti, Massimiliano Stacchini. *Editorial Assistants:* Roberto Marano, Nicoletta Olivanti.

ISSN 1594-7939 (print) ISSN 2281-3950 (online)

Printed by the Printing and Publishing Division of the Bank of Italy

BTP FUTURES AND CASH RELATIONSHIPS: A HIGH FREQUENCY DATA ANALYSIS

by Onofrio Panzarino*, Francesco Potente** and Alfonso Puorro***

Abstract

The paper analyses the interactions between the 'cash' market (MTS Cash) and the futures market (Eurex) of Italian government bonds in terms of liquidity, price correlation and volatility. Based on daily data, the growth of the Eurex market seems to support the tightening of the bid-ask spread of MTS Cash, all things being equal, thus confirming a healthy and efficient link between cash and futures markets. Against this backdrop, a high frequency analysis highlights some episodes of partial divergence between price developments of futures and cash markets, which might be related to differences in the microstructures of the two markets. The futures market is order driven while the cash market is quote driven; furthermore different types of participants are active in each market. At higher frequencies, episodes of unidirectional propagation of volatility shocks from BTP futures to the MTS Cash market materialize, with potential spillovers on cash market liquidity conditions. In this regard, it is also important to consider the role played by High Frequency Traders, whose activity in futures markets may well contribute to explaining the peculiarities in price dynamics highlighted by high frequency data.

JEL Classification: G12, G13, G14

Keywords: market liquidity, HFT, volatility spillover, government bonds.

	Contents	
1.	Introduction	5
2.	Dataset and descriptions of methodologies	8
	2.1 The dataset	8
	2.2 Bivariate GARCH with BEKK specification	10
	2.3 Realized measures: volatility and correlations	11
	2.4 The Granger causality test	12
3.	BTP futures: market developments and contribution to MTS liquidity	13
4.	Correlation analysis	15
5.	Volatility analysis	18
6.	Spillover effects on MTS liquidity conditions	20
7.	Conclusions	21
Ar	nex A. Feasible strategies through futures	22
Re	ferences	24
Та	bles and figures	27

Contents

* Bank of Italy, Market and Payment System Oversight Directorate, e-mail: onofrio.panzarino@bancaditalia.it

^{**} Bank of Italy, Banking Supervision Directorate, e-mail: francesco.potente@bancaditalia.it

^{***} Bank of Italy, Financial Investment Directorate, e-mail: alfonso.puorro@bancaditalia.it

1. Introduction¹

The MTS market, introduced in Italy in 1988, is a regulated market where Italian Government securities are negotiated. It is a quote driven market and it is characterized by the presence of players with market making obligations, committed to quote on a continuous basis on both sides of the market. Participating dealers could conclude transactions only by matching market makers' proposals. An average daily turnover of 4.8 blns was negotiated during 2015 on the MTS, which continues to play a dominant role as an interdealer platform on the Italian Government Bond secondary market.

BTP futures were re-introduced on the German Eurex market on September 14th 2009², when the financial crisis broke the strong correlation among euro denominated sovereign bonds. The Eurex is an order driven market: it allows for direct booking, amendment and cancellation of orders on dealers' negotiation books; transactions derive from the interaction of participating dealers' orders.

Trading volumes negotiated on the BTP futures market were initially extremely low (see Bank of Italy, 2010), but they gradually increased up to a daily turnover in line with other futures on European Government bonds³. Improved liquidity conditions of BTP futures market, which came along with the increase in trading activity, also made BTP futures contracts a very valuable instruments for market making activity, thus significantly strengthening interconnections between BTP futures and MTS Cash market (Bank of Italy, 2015)⁴.

¹ Any views expressed in this paper are the authors and do not necessarily represent those of the Bank of Italy. We thank Claudio Impenna and Roberto Violi for sharing their experience, insights and continuous feedback. Special thanks go to Gioia Cellai and Gaetano Marseglia for their great contribution, helpful discussions and for giving us the time and the opportunity to cooperate and investigate this topic. We also would like to thank anonymous referees for their precious suggestions to the text.

² Between 1991 and 1999, the BTP futures on ten-year maturity was traded on the LIFFE market in London and the Italian Mif. With the introduction of the euro, the convergence of several European countries yields reduced spreads between government bonds, thereby removing the need for different futures for different countries.

³ In terms of open interest, the open positions of BTP, OAT and Bund futures, are respectively equal to 223,714, 188,239 and 1,356,000 (as of 28 April 2015). Furthermore, in April the average number of daily contracts traded amounted respectively to 88,000, 63,000 and about 476,000.

⁴ As an example, on July 20th 2015 the activity on the MTS Cash market deeply suffered from the late opening of the BTP futures market, due to technical problems on the EUREX platform. The future market opened around 10:30 and until then only 15 mln contracts were concluded on the MTS Cash, compared with an average turnover of around 1 bln; average bid-ask spreads was particularly large as well.

In addition, a mature and efficient future market may well contribute to the liquidity of the secondary market by facilitating price discovery. Futures support operational flexibility and allow for a large number of trading strategies that would be hardly viable through cash market only (see Annex A).

Liquidity and efficiency of the MTS Cash market are crucial for a smooth and effective placement of Italian Government bonds. Therefore, it is important to analyze the relationship between the two markets, in a view to assess whether and to what extent the BTP futures market really affects the efficiency of the MTS Cash. The relationships between spot and futures market were widely analyzed in market microstructure studies. Garbade and Silber (1983) show that in commodities markets, in most cases, futures lead spot markets in terms of price discovery and they argue that such relationship is directly affected by liquidity and market size. Furthermore, the authors highlight that the efficiency of the futures market as well as the consistency of prices on the spot and futures market are extremely important for hedging activity. Koontz et al. (1990) also investigate the spatial price discovery mechanism in the livestock market pointing that price discovery process is dynamic and directly influenced by the structure of the market. Other several works investigate this topic and generally support a significant bidirectional information flows between the two markets (see Scalia, 1998) or document a dominant role usually played by futures contracts, above all in commodity markets (Moosa, 2002; Zapata et al., 2005; Fu and Qing, 2006) and treasury bond market as well (Brandt et al., 2007).

More recent studies, based on high frequency data and specifically focused on the BTP futures and MTS Cash markets, confirm the leading role of the futures market compared to the cash market in terms of price discovery. Pelizzon et al. (2014) highlight the impact of credit risk premia on MTS Cash liquidity conditions, while Darbha and Dufour (2015) analyze the implications of liquidity on yield spreads. Rittler (2009), with regard to a completely different market (CO2 emission rights in the EU), shows that - based on high frequency data - the future market leads the spot market in terms of transmission of volatility shocks (volatility discovery). With regard to spillover analysis on other markets, the work of Prasad et al. (2014) has to be mentioned, as well as Conefrey and Cronin (2013) on European bond markets. Such papers are based on a spillover index, first introduced by Diebold and Yilmaz (2009, updated in 2012);

however, such indicator is not easily adaptable to high frequency analyses⁵. Frijns et al. (2013) work elaborates on the causal relationship between an equity volatility index (VIX⁶) and the related future (VXF) within an econometric framework based on Granger Causality test. Such framework was also applied to very high frequency data (15 seconds). This work highlights a bi-directional causality relationship between the two volatility indices.

Frino et al. (2015) highlight the relationship between bid/ask spreads on the equity market, traded volumes and volatility. The methodological framework developed in their paper is the starting point of the analysis carried out in the first part of our Study. Such framework was broadened and adapted so as to include a proxy of the development of the BTP future market as an explanatory variable of bid/ask spreads on BTPs traded on the MTS Cash market.

The results of the first part of the Study, based on daily data, suggest that - in general - the development of the BTP future market may support the compression of MTS bid-ask spreads, with a potentially positive impact on hedging and market making activity. The second part of the Study, based on high frequency data, shows peculiar developments in the correlation between BTP futures and the *cheapest-to-deliver*⁷ bond (as correlation decreases along with increase in data frequency) and a leading role of futures volatility on cash volatility. Such peculiarities, which start to emerge at higher frequency, might reflects different market technological characteristics and different players behavior. In this regard the presence of some specific traders (High Frequency traders - HFT), that operates at such high frequencies, may idiosyncratically affects assets prices dynamics on markets where they operate.

Against this backdrop, it can be argued that the presence of high frequency traders may on its turn affect market efficiency as well. Several analysts highlight that HFTs may positively

⁵ The time series adopted for the estimation of the parameters used for the construction of the indicator are mostly daily or drafted to this frequency, although based on intraday data. At this regard, in our work, we also tried to implement the Diebold and Yilmaz indicator for high frequencies analysis, but it provides no clear guidance. Analysis are underway in order to get a proper fitting of indicator for further research.

⁶ The VIX index is a measure of the implied volatility of the US stock market.

⁷ In general futures contract provides a list of bond that the seller (*short-position*) can deliver to the buyer (*long-position*) at maturity. This list, called *deliverable list*, is usually provided by the market where the futures contract is listed. At the delivery date of a futures contract the agent with the short position in the BTP futures contract has the right to choose the specific bond to deliver and usually the cheapest bond to purchase in the market is most likely to be considered: the *cheapest-to-deliver* (CTD) bond. For the Long-Term Euro-BTP Futures contract the contract terms specify that the underlying instrument is a coupon-bearing debt security issued by the Republic of Italy (BTP), with a remaining life of 8.5 to 11 years and an original maturity no longer than 16 year.

contribute to market liquidity, supporting the compression of bid/offer spreads and reducing transaction costs. In addition, according to some authors, HFTs would contribute to better price discovery (Angel et al., 2010). On the other hand, specialized press as well as non-high frequency traders argue that the most aggressive practices applied by HFTs may even generate situations encompassing a high probability of market inefficiency, aimed at increasing their profit opportunities. In this regard, adverse selection, shadow liquidity, information asymmetries and front running phenomena (Puorro, 2013) are often mentioned among the negative effects of a massive HFT presence. From a different perspective, the recent paper by Caivano (2015) shows that, at least on the Italian equity market, a significant growth in HFTs' activity may bring about a non-negligible increase in yields intraday volatility. Jiang et al. (2014) show that HFTs activity in the US Treasury market may contribute to an increase in volatility in the immediate aftermath of the release of economic news.

In this regard, empirical results of our study highlight risks that - in times of market turmoil - an increase in HFTs activity on the BTP futures market may favor the transmission of volatility spikes to the spot segment, with negative consequences on the liquidity conditions of Italian Government bond secondary market.

Section 2 introduces the data-set employed in the empirical analysis. Section 3 illustrates how the growth of the BTP futures market contributed to MTS market liquidity. Section 4 focuses on correlation developments between selected securities prices traded on the two markets. Section 5 compares volatilities on various frequencies and it analyses volatility spillover risks across markets. Section 6 considers the potential implications of volatility spillover for the liquidity of BTP secondary market, and Section 7 concludes.

2. Dataset and descriptions of methodologies

2.1. The dataset

We rely on two different data-sets, one for a longer term analysis (daily observations) and another for the high-frequency analysis (based on *tick-by-tick* quoting price information). The long-term analysis aims to assess if the introduction of BTP futures market and its development helped in improving the Italian BTP secondary market liquidity conditions. The data-set used includes daily observations covering the time span between 20th September 2006 to 21st august

2015 for: 10-year Italian sovereign bond benchmark *bid-ask* spread, 10-year German Bund yields and the daily turnover traded on MTS Cash and BTP futures market. Table 1 provides main descriptive statistics for above mentioned time series.

High-frequency analysis regards the quoting activity observed on MTS Cash market and BTP futures market (*Eurex*). In order to study their interactions, quoted *Bid* and *Ask* prices are collected from: Long-term (10-year) BTP future contract with maturity September 2015^8 and the related *cheapest-to-deliver* (CTD), that in the period considered in our analysis it turned out to be the BTP 5% March 2025 (shown as *cash 1*). In addition, a similar analysis on two different cash securities with maturity similar to CTD were carried out: BTP 3.75% 09/24 (*cash 2*) and BTP 2.5% 12/24 (*cash 3*), both included in the basket of the deliverable future during the period under review.

The initial dataset consists of intraday tick-by-tick observations regarding 41 contiguous trading days between June 29 until August 28 2015 (an average is calculated when several price changes occur within a single second). A data re-sampling process was subsequently carried out across several time frequencies, from 1 second to 15 minutes. In order to preserve the information content of the data used, in the re-sampling process we always consider the last observation available in the original dataset. In this way we obtain several distinct dataset, one for each frequency considered, and all characterized by equally-spaced observations. The criteria adopted is slightly different from the more common *"fill-in"*⁹ technique usually adopted in case of missing observation or when dealing with not regularly spaced datasets. In place to consider the closest available observation (regardless if the previous or next one available), we employ the last observed one indeed, because (dealing with quoting prices) representative of the effective, still available proposal. Finally, for each business day is considered only the time window that concentrate almost the whole trading activity (from 9:30 AM to 5:00 PM). Table 2 presents summary statistics for the *log-returns* of the considered assets.

⁸ IKU5 instrument on Bloomberg. Italian government bond futures are quoted on four different delivery months (March, June, September and December) and are written on a theoretical 10 year bond with a coupon of 6%; each contract has a notional value of $100.000 \in$ and a minimum price change (tick) of $0.01 \in$; therefore one contract position exposes the owner to +/- 10 \in changes in values for every price tick.

⁹ The "fill-in" technique is largely adopted in literature in order to deal with missing data in time series analysis (see Girardi and Impenna, 2013; Upper and Werner, 2002).

In the following sections we describe the econometric tools used to investigate correlations, volatilities and liquidity transmissions in the Italian *spot* and *futures* markets.

2.2. Bivariate GARCH with BEKK specification

Economic literature developed a large variety of univariate GARCH-type models that properly capture time series volatility dynamics. Because more interested in modeling the mutual relationship between two markets, we adopt a multivariate GARCH model. In such a way we had the possibility to simultaneously estimate both volatility and correlation of the two considered time series. In the context of VECH-like¹⁰ model we employ the GARCH-BEKK specification introduced by Engle and Kroner (1995) and largely adopted in literature because of the advantage of a relatively small number of parameters that have to be estimated (11 in place of 21 for the bivariate case). However we decide to not employ a multivariate GARCH model for datasets with observation collected at frequencies above 5 minutes. This also to consider the great impact of the microstructure noise that arises at such high sampling frequencies. The BEKK representation describes the conditional covariance dynamic using the following (2 × 2) matrix structure:

$$H_t = C \cdot C' + \sum_{i=1}^m A_i (\varepsilon_{t-i} \varepsilon'_{t-i}) A'_i + \sum_{j=1}^s B_j H_{t-j} B'_j \qquad (1)$$

where A_i , B_j and C are (2 × 2) coefficient matrices; C is an upper triangular matrix while A_i e B_j are symmetric and positive definite. This model specification automatically guarantees the positive definiteness of the conditional covariance matrix H_t and allows to dynamically captures the mutual interdependence between the volatility series. In particular, while the non-diagonal elements of matrix B measure the extent to which the conditional variance of one market is correlated with the lagged variance of the other market, the diagonal elements depict the effects on current squared volatilities caused from their own past. In case of a GARCH-BEKK(1,1) the general functional form expressed in (1) reduces to:

$$H_t = C \cdot C' + A(\varepsilon_{t-i}\varepsilon'_{t-i})A' + BH_{t-i}B' \qquad (2)$$

¹⁰ The diagonal representation of the VECH model (see Bollerslev et al., 1988) requires a large number of parameters to be estimated (21 in the bivariate case).

with

$$A = \begin{bmatrix} a_{11} & a_{12} \\ a_{21} & a_{22} \end{bmatrix}, \quad B = \begin{bmatrix} b_{11} & b_{12} \\ b_{21} & b_{22} \end{bmatrix} \text{ and } C = \begin{bmatrix} c_{11} & c_{12} \\ 0 & c_{22} \end{bmatrix}$$

2.3. Realized Measures: Volatility and Correlations

In recent time the volatility literature has steadily progressed toward the use of higherfrequency data and recent works has clarified the comparative desirability of alternative volatility estimators. The so-called *realized measures* are placed in this context. Andersen and Bollerslev (1998) show that, under the usual assumption that log asset prices evolve as a diffusion process, *realized volatility* and *correlation* computed from high-frequency intraday returns are effectively error-free¹¹ measures. Moreover, such estimators are easily constructed (respectively sum of intra-period high-frequency squared returns or cross products returns), they allow to limit the effect of possible market microstructure noise effects (Andersen et al., 2001, 2003) and they are recognized as being the closest estimate of the true latent integrated volatility (Andersen et al., 2005). The first quantity we computed is the *Realized Variance (RV)* measure, formally defined through the following formula:

$$RV_t^f = \sum_{n=1}^{N^f} r_{n,t}^2$$
 (3)

where RV_t^f is the *Realized Variance* referred to the (t - 1, t) time interval, $r_{n,t}^2$ is the n^{th} squared logarithmic price variation observed at the considered frequency f and N^f is the total number of the observation that falls in each time interval (in our case a 5-minute time interval is adopted). We iterate this procedure for six sampling frequencies (f = 1 sec, 3 sec, 5 sec, 10 sec, 15 sec, 30 sec). Afterwards a *Realized Volatility* (*RVol*) measure is simply computed as the square root of the abovementioned *Realized Variance* and expressed in daily percentage points using the following formula:

$$RVol_t^f = 100 \cdot \sqrt{day^f \cdot RV_t^f} \qquad (4)$$

¹¹ Estimated realized volatilities and correlations are "error-free" in the sense to result approximately not affected by measurement error (see Andersen and Bollerslev, 1998).

where the day^{f} coefficient indicates the number of daily observations at the sampling frequency *f*. Similar methodologies as previously described are subsequently used to compute the following *Realized Covariance (Rcov)* e *Realized Correlation (RCor)* measures:

$$RCov_t^f = \sum_{n=1}^{N^f} r'_{n,t} \cdot r''_{n,t} \qquad RCor_t^f = \frac{RCov_t^f}{\sqrt{RV'_t^f \cdot RVI''_t^f}} \qquad (5)$$

where the $r'_{n,t}$ and $r''_{n,t}$ are the logarithmic price variations observed on two different assets. Afterwards a *kernel approach* is employed in order to smoothly estimate the probability density functions of the above mentioned quantities. In our case, for each considered sampling frequency of the considered *Realized Correlation* ($RCor_t^f$) and *Realized Volatility* ($RVol_t^f$) measures, the *kernel density estimator* are:

$$F_{RCor_{t}^{f}}(x) = \frac{1}{Th} \cdot \sum_{t=1}^{T} k\left(\frac{x - RCor_{t}^{f}}{h}\right), \quad F_{RVol_{t}^{f}}(x) = \frac{1}{Th} \cdot \sum_{t=1}^{T} k\left(\frac{x - RVol_{t}^{f}}{h}\right)$$
(6)

where *T* is the length of the considered time series, *h* is the *bandwidth* (a parameter that controls the smoothness of the resulting probability density curve and it is usually chosen according to some rules, in order to produce reasonably smooth densities) and $k(\cdot)$ is the *Gaussian kernel function* defined with the following formula:

$$k(x) = \frac{1}{\sqrt{2\pi}} e^{-\frac{1}{2}x^2}$$
 (7)

2.4. The Granger Causality Test

We perform the *volatility spillover* analysis in an attempt to identify potential transmission effects between the two considered markets, employing a sequence of *Granger Causality Tests* (Granger, 1969). Such econometric tool, largely adopted in literature, assumes a time series *X* to "cause" (in the sense of Granger) another series *Y*, if the former is useful in forecasting the latter within a given time lag. More rigorously, Granger causality concept is defined as follows:

$$X \xrightarrow{G-cause} Y \Leftrightarrow \mathbb{E}(y_t | y_{t-1}, y_{t-2}, \dots, x_{t-1}, x_{t-2}, \dots) \neq \mathbb{E}(y_t | y_{t-1}, y_{t-2}, \dots)$$
(8)

If a bivariate VAR model of order p is employed to describe $y_t e x_t$ dynamics, the Gcausality condition recalled in (8) may be tested relying on the following joint hypothesis:

$H_0: \boldsymbol{\beta}_1 = \boldsymbol{\beta}_2 = \cdots = \boldsymbol{\beta}_p = \boldsymbol{0} \qquad (9)$

where the β_i coefficients capture the cross-interactions between y_t and the lagged values of x_t . If the null hypothesis holds, we can exclude that X Granger-cause Y. The inference on such parameters happens through a Test statistic that, under the null hypothesis, follows a *Fisher* distribution with p and T - 2p - 1 degrees of freedom (where p is the order, or lag, adopted in the VAR specification and T is the time series length). Significantly large values of the above mentioned F-Statistic may allow to reject the null hypothesis, reported in (9), detecting the presence of a causal relationship (in the sense of Granger) between the two series.

Finally, before conducting Granger causality tests we check across all the sampling frequencies if the considered time series are stationary. We adopt the *Augmented Dickey Fuller* (*ADF*) test in order to detect the presence of unit root. Resulted ADF statistics are reported in Table 3 and suggests that we can strongly reject the presence of a unit root in all considered time series, that means we can assume all series be stationary across all frequencies.

3. BTP futures: market developments and contribution to MTS liquidity

The implications in terms of liquidity for MTS market arising from the development of BTP futures were evaluated adopting an econometric analysis. The starting point is represented by a simple functional specification that relates the equities bid-ask spread to trading volumes and volatility (Frino et al., 2015). In particular, a functional relationship between the logarithm of the bid-ask spread on the generic ten-year BTP (dependent variable) and the logarithms of the traded volumes on the MTS and the volatility of the return of the ten-year BTP has been tested as part of a simple OLS scheme (calculated on a monthly floating window).

The results related to the functional specification, not presented in the text, show a direct relationship between increased volatility and the widening of the bid-ask spreads, where an increase in the traded volumes on the MTS favors spread tightening. However, the Durbin Watson¹² statistics calculated on this specification suggest the presence of a strong positive

 $^{^{12}}$ The Durbin-Watson is a statistical test used to detect the presence of residual autocorrelation in a regression analysis. The value of the statistic is always between 0 and 4; a value of 2 indicates presence of autocorrelation while small index values (high) show positive correlation (negative) between residues.

auto-correlation in the error terms, as also indicated by the residual correlogram¹³. In addition, a very low R^2 value shows little explanatory power, thus suggesting an extension to the mentioned specification.

In order to assess the contribution to the MTS liquidity related to the development of the futures market, the functional specification has therefore been enriched by the inclusion of other explanatory variables. In particular, the daily traded volume of the derivative was introduced, as a proxy of the BTP futures market size. In addition, the spread between the BTP ten year yield and the yield of Bund with similar maturity was added. It can be argued that an increase in risk premium required by investors to hold BTPs (as measured by the BTP-Bund spread) may also negatively reflect on their liquidity (Pelizzon et al., 2014). Finally, the functional specification has been extended with some lag terms of the bid-ask spread in an attempt to reduce the problem of auto-correlation of the error term. Results are reported in Table 4.

The analysis seems to support the idea that the increase in the size of BTP futures market may positively affect the liquidity of MTS Cash market. On the other hand, the widening in BTP-Bund spread (and the related increase in risk premium) may adversely affect the liquidity of the BTP. Furthermore, according to the Durbin Watson statistic no problems of autocorrelation among lags of the error term seem to emerge (see also the correlogram reported in Figure 1).

The value of R^2 shows a not negligible explanatory power of the proposed specification. In order to take into account potential problems of multi-collinearity arising from the strong correlation among the increase in MTS traded volumes and the increase in futures traded volumes, a two-steps regression procedure has been carried out. In the first step, the futures traded volume has been regressed against the MTS traded volume. The errors of the first step regressions have been used as regressor instead of the futures volumes in the functional form; results are reported in Table 5.

The analysis seems to confirm empirically the positive effect of the development of the BTP futures market on the liquidity of MTS Cash market in terms of tightening of bid-ask spread,

¹³ The correlogram is a representation technique that depicts the value of the autocorrelation of a time series reported with different delays (x-axis). From an inferential point of view, the represented horizontal stripes allow to identify the confidence interval (95%) of acceptance of the hypothesis of correlation values equal to 0. Therefore, in a nutshell, the out of band autocorrelations are to be considered as "statistically significant. "

other being equal. Therefore, the development of the derivative market seems to have actually supported the liquidity of the secondary, cash market. In the following paragraphs an high frequency analysis is carried out in order to analyze more thoroughly the interconnections between the two markets in terms of correlation and propagation of volatility shocks.

4. Correlation analysis

The correlation analysis between BTP futures and CTD returns has been first of all carried out through the estimation of a bivariate GARCH model as described in Section 2.

The graphs outlined in Figure 2 show episodes of correlations breaks which are promptly reabsorbed. Analyzes based on GARCH estimates do not seem reliable for higher frequencies (Andersen, 2005). Therefore, specific analysis were adopted on the basis of descriptive statistics. In addition, for higher frequencies, in order to verify that the behavior of futures-CTD correlations does not represent a "mechanical" effect of the increased detection of frequencies¹⁴, an analysis of the dynamics of the correlation of pairs of cash securities with the same maturity was also conducted.

Plots reported in Figure 3 show the correlations dynamics at different sampling frequency between the BTP futures and its relative CTD bond (*cash 1*). For each trading day, the correlation between log-return of pairs of above mentioned securities was calculated, allowing to carry out a cross-sectional analysis of the sample estimates. This procedure was re-iterated for each sampling frequency. Figures show that the average correlation between futures and CTD is close to one for lower frequencies, as well as the correlation between cash securities with similar characteristics.

Therefore, as the sampling frequency increases, a progressive reduction of the correlation for all pairs of titles analyzed is observed on average. However, this effect seems more emphasized for the futures-CTD couple that, at least in theory, should be the most similar pair of titles. In particular, at the 1 second frequency the average correlation between futures and CTD is lower (about 30%) than the correlation of the other couples of cash securities.

¹⁴ Increasing the frequency of detection, even minimum time discrepancies between quoted prices of securities on different markets are more likely, thus showing a reduced correlation.

This phenomenon is more evident if we exclude from the data set the days when auctions of ten years BTP are carried out (June 30, July 13, July 30 and August 28; see Figure 4). At this regards, the price dynamics in the moments immediately before and after the auctions could reflect idiosyncratic behaviors due to changes in market makers inventories. Those adjustments may produce a greater impact on the cash market, thus affecting the correlation between pairs of cash securities.

However, computing daily correlations based on high frequency data may significantly suffer from outliers observations. Even in large samples, it can't be excluded that few observations may influence the overall correlation estimate if values are considerably divergent from others. In our case, correlations estimated in datasets with a sampling frequency below 10 seconds may potentially overweight atypical price dynamics observed in tight time intervals, for example if pretty instantaneous bid-ask widening shocks occurs. We thus decide to make use of realized measures mentioned in Section 2, as a straightforward way to properly cope with such technical drawbacks. In this regard, intraday realized correlations dynamics are captured at four distinct sampling frequencies and kernel density estimators are adopted in order to deal with outliers issues. Results are shown in Figure 5 and support what emerged from the previous box plots.

The partially divergent trends between futures and CTD prices arising at high frequencies may be explained by differences in some structural features of the MTS and futures markets that contribute to shape the price discovery processes across the two trading venues. First of all, the cash market is limited to market makers, while the futures market allows a wider range of interested parties to have access to the trading platforms. Second, the order-driven nature of the futures market makes it easier for all operators to enter, delete or modify their trading orders on a continuous basis. Moreover, futures contracts allow to implement a large deal of trading strategies (see Annex A). Finally, this operational flexibility of the futures market, available to all categories of investors, has encouraged the use of automated trading systems (*algo-trading*); some of these systems – known as High Frequency Trading or HFT – are characterized by high computational capacity and very advanced technologies and are able to carry out trades in extremely short times.

High-frequency traders initially developed in the US stock market about ten years ago, thanks to a favorable regulatory environment¹⁵ and a market microstructure¹⁶ that favored their developments. The futures markets in the US were immediately subject to the attention of HFTs. For example, the presence of HFTs on the S&P 500 futures market is significant (30-40%) as well as on other US traded asset class futures markets, such as oil or government securities. A recent study published by the Federal Reserve Bank of New York¹⁷ shows that more than half of trading of US Treasuries are carried out by automated trading systems and a significant part of those trades is carried adopting high-frequency technologies. In Europe, numerous researches show a non-negligible presence of HFTs on several European equity markets, where HFTs seem to be responsible for a trading percentage between 20% and 40% of the total. Over the past two years, the increased use of electronic trading platforms and the increased fragmentation of the markets are slowly encouraging HFT colonization of other markets such as foreign exchange and fixed income.

Although official statistics about the presence of high-frequency traders in the European fixed income futures markets are not present, the order-driven nature of such markets and their growing liquidity are undeniable attraction factors for high-frequency traders. Therefore it is reasonable to assume a significant presence of these operators also in the BTP futures market. This could explain the peculiar dynamics of the futures price that, at high frequencies and for short time intervals, may partially deviate from the dynamics of the price of the underlying bond, with implications in terms of reduction of the correlation observed. In addition, negotiations on the BTP futures are realized on a platform technologically different and physically separate from the platform on which negotiations on BTP cash are carried out. Therefore, any hedging or arbitrage operation that involves both markets and that in turn exerts a pressure to prices realignment, necessarily it requires some technical implementation time. The realization of an arbitrage strategy involving securities traded on the same platform would undoubtedly be faster.

¹⁵ The decisive impulse to the development of high frequency was offered by Regulation National Market System (NMS Regulation) of 2005; European innovations contained in the US standard were later introduced by MiFID.

¹⁶ One of the first stimulus to the birth of HTT was offered by the decision of the Securities and Exchange Commission that, in the early 90s, allowed the use of Electronic Communications Networks (ECN) as electronic trading systems alternatives to regulated markets.

¹⁷ "Automated Trading in Treasury Markets"; TMPG Consultative paper, October 2015.

5. Volatility analysis

Studying interdependence in financial markets has increasingly attracted significant attention in recent times and the vastly growing literature in this topic provides strong evidence in this regard. Against this backdrop, we focus our analysis on financial markets volatility cross-interactions. Volatility is a key elements in financial markets analysis, ranging from asset allocation to risk management needs. Our analysis is aimed to detect to what extent futures market volatility might play a leading role *vis-a-vis* the cash market volatility and on its turn might affect BTP secondary market liquidity. The analysis leverages on several econometric tools outlined in Section 2. We first consider a bivariate GARCH specification; Figure 6 illustrates the estimated conditional volatilities dynamics of the BTP futures contracts and its relative *cheapest-to-deliver*.

The two series show similar intraday volatility patterns through the considered time period with frequent *spikes* episodes and, as the data frequency increases, more structural breaks occurs in the volatility patterns for both considered series. In order to deepen the analysis considering the higher sampling frequency dataset we move to the so called *realized measures*, which are recognized as a more accurate volatility measure with better forecasting capabilities in high-frequency environments.

As for the correlation analysis, probability density estimates are computed for time series volatilities as well, by employing the Kernel methodology outlined in Section 2. Results are showed in Figure 7 and confirms the greater volatility detected on BTP futures instruments compared with respect to all bonds in the deliverable basket. The higher volatility is partially explained by the optionality enclosed in the derivative contracts that exposed the *long* position to buy, at maturity, the underlying asset that has performed worse (*delivery option*); therefore, compared to related underlying asset, BTP futures prices pay a premium to face such risk exposure.

Now therefore we investigate potential transmission mechanisms between the *futures* and the *spot* market in order to mainly asses if the higher volatility recorded on the derivative contract may potentially propagate on the Italian Government Bond secondary market also affecting its liquidity conditions.

In order to investigate such spillover phenomena across the two markets, we leverage on the concept of Granger Causality test for determining whether one time series is useful in forecasting another. In a first step we focus on volatility transmission effects. For each trading day and for each considered sampling frequency we estimate a *lead-lag* VAR framework using absolute *log-price* variations of BTP futures and related CTD contract as proxies for their respective volatility dynamics¹⁸. Figure 8 reports Granger causality test statistics and a common box-plot representation is adopted in order to better represents all values obtained across different frequencies and trading days included in our sample period.

Results provide a strong statistical evidence for a causal relationship running from the future market to the spot market with a tendency to strengthening as the sampling frequency increases¹⁹. For the sake of completeness we also investigate the opposite causal effect (from *cash* to *future*) with much weaker results than vice versa.

Figure 8 highlights that for lower sampling frequency observations (15 to 5 minutes), Granger Causality test fails to be conclusive with a 99% confidence intervals (*dashed line*) on several trading days, implying a bidirectional transmission mechanisms between the two markets; on the other hand the F-statistics values above the threshold obtained for the ultrahigh-frequency datasets support a strong unidirectional spillover effect only from the futures to the cash market.

Showed results seem to point out that to better understand causal relationship on such liquid markets, it requires to gather the much richer information enclosed in the higher sampling frequency datasets. In our case in order to properly assess a causal relationship between the two markets and identify unidirectional spillover we had to fall under the five minutes sampling frequency. Such empirical evidence may be strictly linked to the peculiar microstructure characteristics of the BTP futures market that we have highlighted in Section 4 and that potentially have non negligible implications for the MTS Cash market as well.

¹⁸ The number of lags is selected according the Bayesian information criteria (BIC).

¹⁹ A recent work (Rittler, 2009), based on high-frequency data, is devoted to analyze the relationship of European Union Allowance *spot* and *futures* prices; the work reaches similar conclusions. The paper analyses both a price discovery process (Information Share - Hasbrouck 1995) and a volatility spillover analysis between the two markets (employing a Granger Causality test and a multivariate GARCH specification). Authors detect unidirectional volatility transmission from the futures to the spot market at highest frequencies.

A detailed analysis referred to the US Treasury market (conducted by U.S. Department of the Treasury, Federal Reserve System and Bank of New York, the U.S. Securities and Exchange Commission and the Commodity Futures and Trading Commission²⁰) states that in normal condition, the link between cash and futures market positively contributes to an efficient prices formation process; however such relation between the two markets may, in same circumstances, constitute a channel for shocks transmission from one market to others. In the same report a cross-market analysis states that top ten proprietary traders activities usually begins on the futures market and subsequently shifts on the cash market in a five milliseconds time interval.

6. Spillovers effects on MTS liquidity conditions

As shown in Section 3, the development of the futures market seems to support the liquidity of the MTS market. However, propagation of volatility from the futures to the cash market could have a negative impact on the MTS liquidity (as measured by bid-ask spreads²¹). Market makers protect themselves from increased volatility by reducing the price at which they are willing to buy and increasing the price at which they are willing to sell, with a consequent increase in the bid-ask spread. Therefore, the increase in volatility on the cash market naturally leads to a deterioration in liquidity conditions²². In order to empirically test the relationship between volatility and bid-ask spreads, appropriate tests of Granger Causality were conducted for different frequencies and for each trading day.

Figure 9 shows the presence of a causal relationship between market volatility and bid-ask spreads quoted by market makers, especially for frequencies below 30 seconds. Moreover the trend of the F-statistic, above the limit threshold, with a maximum "local" value around ten seconds²³, suggests the possibility that the widening of the bid-ask spreads occurs for more

²⁰ "The US Treasury market on October 15, 2014"; Joint staff report, October 2014.

²¹ Bid-ask spreads represents only one dimension of market liquidity. However in the purpose of our analysis it may be considered as the most relevant one. A widening in the bid-ask spreads represents higher transaction costs regardless of the orders size. This is not always true for other market liquidity measures such as depths and turnovers. Moreover the binding nature of the quoted proposal submitted on the MTS Cash platform makes bid-ask spreads detected on this market highly informative of its liquidity condition.

²² Moreover, it is not inconceivable that market-makers directly infer market volatility from exchanged futures prices in order to adjust their proposal on the cash market, as the futures market, characterized by more trades than those recorded on a single cash security, can offer more frequent and granular information.

²³ For 10 and 5 seconds frequencies, Granger test is conclusive for 70% of analyzed trading days.

persistent volatility spikes, presumably in order to avoid the noise of volatility meddle with their proposals.

7. Conclusions

A liquid futures market contributes to the efficiency and proper functioning of the underlying cash market, favoring price discovery and offering market players the possibility to more easily hedge their positions. In addition, a future market characterized by an adequate soundness facilitates market makers activity; it promotes the compression of bid-ask spreads, and therefore tends to improve the underlying liquidity.

The futures and the underlying cash market are strongly intertwined. However, for higher frequency observations (seconds), price dynamics tend partially to diverge, including those between futures and CTD. This may be due to differences in micro-structure of the two markets and to different players active in each market. Given the order-driven nature of BTP futures market, it is reasonable to assume that the negotiation of this instrument may also be affected by the activity of high frequency traders (HFT) likewise what occurred in recent years on equity order-driven markets.

The analysis highlights that significant changes in volatility occurring on the futures market tend to spread to the cash market, thus affecting the liquidity of the latter (Bank of Italy, 2015): it is indeed fully rational for market makers to adapt their offers when the increase in volatility is deemed not negligible and not transitory. The cash market volatility dependence from the futures volatility should not be underestimated, however. It is also possible that, when market conditions are fragile and particularly sensitive to any negative news, the presence of HFT operators in the futures market, may amplify futures prices volatility thus magnifying the temporary liquidity drop in the cash market.

Annex A. Feasible strategies through futures

Futures contracts may be used to implement several trading strategies. In more details:

- a) Position Hedging: futures are often used to minimize risk exposure on specific fixed income positions. This feature proved to be particularly useful for market makers, also in the light of more recent regulatory developments. According to a recent BIS study (BIS, 2014), many market makers have reduced their inventories of securities as a response to increased regulatory constraints, which potentially weakens their ability to absorb large sale and purchase orders. A futures market characterized by an high degree of liquidity allows market makers to effectively manage their book even without large inventories.
- b) Portfolio Hedging: futures are an effective tool also to hedge entire bond portfolios (Moorad and Choudry, 2004). The adoption of BTP futures, along with German Bund futures, might improve the hedge effectiveness of diversified portfolios of fixed income securities in respect to the hedge effectiveness brought about with the use of Bund futures alone (Bessler et al., 2014). Such a usage of BTP futures has become even more common in the aftermaths of the European government debt crisis in 2010 2011 when, on one hand, bonds characterized by larger risk premiums, including BTPs, assumed price dynamics strongly similar, while, on the other hand, the correlation decreased against the Bund, and, to a lesser extent, the OAT. Therefore, the use of BTP futures increased as hedging instruments for positions not only in Italian government bonds.
- c) Arbitrage: considerable discrepancies between the futures price and the price of the cheapest-to-deliver determine relative value²⁴ opportunities that can be seized by purchasing (or selling) the futures and simultaneously selling (or buying) the CTD (so called "basis trading"). This acts as a natural pressure and price discrepancies are rapidly reabsorbed. In this sense, the arbitrage activities, facilitated by the availability of a liquid futures market, contributes to the price discovery also on cash instruments.

²⁴Although many consider the basis trading an arbitrage opportunity, it is subject to a modest risk. For a detailed description of the basis trading (see Burghardt and Belton, 1994).

d) Speculation: futures contracts stimulate speculative traders activity, because, unlike what happens in cash markets, they can easily take also short positions. Although speculation is often linked to aggressive or opportunistic behaviors, the presence of an adequate number of speculators operating in both directions can help improve price discovery, thus increasing market efficiency.

In conclusion, the development of BTP futures market may represent, ceteris paribus, a factor supporting cash market liquidity, as it can result in a reduction of market making activity costs (position/portfolio hedging). Moreover, the possibility offered by the futures to quickly and efficiently implement (with limited use of cash) directional strategies ("speculation") or arbitrage operations contributes to improve price discovery.

References

Andersen, T., Bollerslev, T. (1998), "Answering the Skeptics: Yes, Standard Volatility Models Do Provide Accurate Forecasts", International Economic Review, 39, 885-905.

Andersen, T., Bollerslev, T., Diebold, F.X. and Ebens, H., 2001, "The distribution of realized stock return volatility", Journal of Financial Economics 61, 43-76.

Andersen, T., Bollerslev, T., Diebold, F., X., and Labys, P., 2003, "Modeling and Forecasting Realized Volatility", Econometrica 71, 529-626.

Andersen, T., Bollerslev, T. and Meddahi, N., 2005, "Correcting the errors: Volatility Forecast evaluation using high-frequency data and Realized volatilities", Econometrica.

Angel, J., Harris, L., and Spatt, C. S., 2010, "Equity Trading in the 21st Century".

Bank of Italy, 2010, "Relazione Annuale".

Bank of Italy, April 2015, "Financial Stability Report".

Bessler, W., Leonhardt, A., and Wolff, D., April 2014, "Analyzing Hedging Strategies for Fixed Income Portfolios: A Bayesian Approach for Model Selection".

BIS, 1998, "How Resilient are Financial Markets to Stress? Bund Futures and Bonds during the 1998 Turbulence", BIS Papers, No. 12, 110-123.

BIS, November 2014, "Market Making and Proprietary trading: Industry Trends, Drivers and Policy Implications"; CGFS Papers.

Bollerslev, T., Engle, R. and Wooldridge, J., 1988, "A capital asset pricing model with time-varing covariances", Journal of Political Economy 96 (1): 116-131.

Brandt M., Kavajecz, K., and Underwood, S., 2007, "Price discovery in the treasury futures market", The Journal of Futures Markets, Vol. 27, No. 11, 1021–1051.

Burghardt, G., and Belton, T., 1994, "The Treasury Bond Basis: An in-Depth Analysis for Hedgers, Speculators, and Arbitrageurs", McGraw-Hill, Library of Investment and Finance.

Caivano, V., March 2015, "The impact of high-frequency trading on volatility", Consob, Quaderni di Finanza.

Choudry, M., 2004, "Using Bond Futures Contracts for trading and Hedging".

Conefrey, T., and Cronin, D., 2013, "Spillover in Euro Area Sovereign Bond Markets", Central Bank of Ireland, Research Technical Paper.

Darbha, M., Dufour, A., 2015, "Euro Area Government Bond Market Liquidity".

Diebold, F., X., and Yilmaz, K., 2009, "Measuring Financial Asset Return and Volatility Spillovers, with Application to Global Equity Markets", Economic Journal, 119.

Diebold, F., X., and Yilmaz, K., 2012, "Better to Give than to Receive: Predictive Directional Measurement of Volatility Spillovers", International journal of Forecasting.

Engle, R. and Kroner, K, 1995, "Multivariate simultaneous Generalised ARCH", Econometric Theory, 11, 122-150.

Frino, A., Mollica, V., and Zhang, S., May 2015, "Determinants of Bid-Ask Spreads in Time-Series Analysis: A Note".

Frijns, B., Tourani-Rad, A., and Webb, R., 2013, "On the Intraday Relation between the VIX and its Futures".

Fu, L.Q. and Qing, Z.J., 2006, "Price discovery and volatility spillovers: Evidence from Chinese spot-futures markets".

Garbade, K.D., and Silber, L.S., 1983, "Price Movements and price discovery in futures and cash market", The MIT Press, Vol. 65 No. 2, 289-297.

Girardi, A., and Impenna, C., 2013, "Price Discovery in the Italian Sovereign Market: The Role of Order Flow", Banca d'Italia, Temi di discussione n. 906.

Granger, C, 1969, "Investigating causal relations by econometric models and cross-spectral methods", Econometrica 37: 424-438.

Jiang, G. J., Lo, I., and Valente, G., 2014, "High-Frequency Trading around Macroeconomic News Announcements: Evidence from the U.S. Treasury Market", Bank of Canada, Working Paper n. 56.

Koontz, S.R., P. Gracica and Hudson, M.A., 1990, "Estimation and hypothesis testing of cointegration vectors in Gaussian Vector Autoregressive Models", Econometrica 59.

Moosa, I.A., 2002, "Price discovery and risk transfer in the crude oil futures markets: structural time series evidence", Banca Monte dei Paschi di Siena SpA, Vol. 31, Blackwell Publishers.

Pelizzon, L., Subrahmanyam, M., Tomio, D., Uno, J., 2014, "Limits to Arbitrage in Sovereign Bonds Price and Liquidity Discovery in High-Frequency Quote-Driven Markets"

Prasad, N., Grant, A., R., and Suk-Joong Kim, 2014, "Time Varying Volatility Indexes and Their Determinants: Evidence from Developed and Emerging Stock Markets".

Rittler, D., 2009, "Price Discovery, Causality and Volatility Spillovers in European Union Allowance Phase II: A High Frequency Analysis", Heidelberg University Press.

Scalia, A., 1998, "Information transmission and causality in the Italian Treasury bond market", Journal of Empirical Finance, n.5 361-384.

TMPG, 2015, "Automated Trading in Treasury Markets", TMPG Consultative paper.

Upper, C., and Werner, T., 2002, "Time variation in the tail behaviour of bund futures returns", ECB, Working paper n. 199.

Zapata, H., Fortenbery, T.R. and Armstrong, D., 2005, "Price discovery in the world sugar futures and cash markets: implications for the Dominican Republic", Staff Working Paper No.469, Agricultural and Applied Economics.

Table n.1

frequency	# Obs	BID - ASK 10Y (bp)		Spread BTP - BUND (bp)		BTP daily volatility (bp) (diff yield, rolling 30 days)		Daily turnover MTS (mln)		Daily turnover BTP futures (mln)	
		Mean	Stand. Dev	Mean	Stand. Dev	Mean	Stand. Dev	Mean	Stand. Dev	Mean	Stand. Dev
daily	2328	1.5	1.13	167	123	6	4	4.000	2.180	3.155	3.571

Table n.2

Prices (%, logarithmic return)										
frequency	# Obs.	Future		BTP 5% marzo 2025 (ctd - cash1)		BTP 3.75% 9	9/2024 (cash2)	BTP 2,5% 12/2024 (cash3)		
		Mean	Stand. Dev	Mean	Stand. Dev	Mean	Stand. Dev	Mean	Stand. Dev	
15 min	1472	0.00001478	0.06642218	0.00007954	0.06433042	- 0.00021828	0.06303222	- 0.00041157	0.06679181	
10 min	2208	0.00000985	0.05560975	0.00005303	0.05429722	- 0.00014552	0.05311328	- 0.00027438	0.05784421	
5 min	4416	0.00000493	0.03999072	0.00002651	0.03917962	- 0.00007276	0.03790682	- 0.00013719	0.04136402	
1 min	22080	0.00000099	0.01894237	0.00000530	0.01926481	- 0.00001455	0.01880501	- 0.00002744	0.01892717	
30 sec	44160	0.00000049	0.01351648	0.00000265	0.01371133	- 0.00000728	0.01291285	- 0.00001372	0.01313673	
15 sec	88320	0.00000025	0.00963156	0.00000133	0.01056484	- 0.00000364	0.00891502	- 0.00000686	0.00919920	
10 sec	132480	0.00000016	0.00782027	0.0000088	0.00863813	- 0.00000243	0.00724343	- 0.00000457	0.00781955	
5 sec	264960	0.0000008	0.00550670	0.00000044	0.00636090	- 0.00000121	0.00549775	- 0.00000229	0.00555523	
4 sec	331200	0.0000007	0.00492509	0.0000035	0.00592192	- 0.00000097	0.00503454	- 0.00000183	0.00493525	
3 sec	441600	0.00000005	0.00428743	0.00000027	0.00524915	- 0.0000073	0.00436821	- 0.00000137	0.00430648	
2 sec	662400	0.0000003	0.00348571	0.00000018	0.00437780	- 0.00000049	0.00358964	- 0.00000091	0.00355369	
1 sec	1324800	0.00000002	0.00244672	0.00000009	0.00323978	- 0.0000024	0.00255557	- 0.00000046	0.00254182	

Table n.3: Unitroot test for stationarity

fraguancy	Future	CTD	spread bid-ask	1% significance	
Jrequency	(abs. returns)	(abs. returns)	(abs. returns)	level	
15 min	-3.872	-3.541	-5.013	-4.197	
10 min	-4.841	-5.202	-6.987	-3.968	
5 min	-7.674	-7.964	-6.942	-3.665	
1 min	-11.862	-10.635	-11.273	-3.472	
30 sec	-13.326	-12.013	-15.086	-3.452	
15 sec	-12.833	-13.634	-19.246	-3.444	
10 sec	-14.981	-14.400	-22.170	-3.438	
5 sec	-18.296	-20.236	-29.293	-3.434	
4 sec	-20.429	-22.428	-32.253	-3.434	
3 sec	-22.987	-24.876	-38.101	-3.434	
2 sec	-28.699	-32.539	-44.557	-3.433	
1 sec	-41.493	-49.481	-64.179	-3.433	

(Augmented Dicky-Fuller)

Table shows for each sampling frequency the averaged ADF test statistics across all trading days. Values reported are in general beyond the 1% confidence level and permit to reject the hypothesis of a unit root presence, giving evidence of a stationarity series (the more negative the ADF statistics is, the stronger the rejection of the hypothesis that there is a unit root at some level of confidence).

Table n.4: BTP futures market development and contribution to MTS liquidity

OLS estimates, daily data from 2009/09/15 to 2015/08/21 Dependent Variable: l_BID_ASK_10Y Heteroscedasticity-consistent standard errors

	Coefficient	Std.err	ror	t-statistic	p-value	
Const	0.29804	0.1626	33	1.8326	0.06706	*
MTS_Volumes	-0.0543848	0.02448	349	-2.2212	0.02649	**
BTP_Volatility	0.0163743	0.02561	154	0.6392	0.52276	
Future_Volumes	-0.033391	0.01495	557	-2.2327	0.02571	**
Spread_IT_DE	0.0200428	0.01202	266	1.6665	0.09581	*
BID_ASK_10Y_(1)	0.501936	0.06587	738	7.6197	< 0.00001	***
BID_ASK_10Y_(2)	0.111692	0.06185	559	1.8057	0.07116	*
BID_ASK_10Y_(3)	0.0692628	0.06328	353	1.0945	0.27393	
BID_ASK_10Y_(4)	0.214281	0.06039	932	3.5481	0.00040	***
R ²	0.8	81801	Adj R ²			0.881187
F(8, 1540)	147	4.753	P-value(F)			0.000000
			Durbin-W		2.047764	

Figure n.1: Correlogram

(auto-correlation and partial auto-correlation of the residuals)



Table n.5: BTP futures market development and contribution to MTS liquidity OLS estimates, daily data from 2009/09/15 to 2015/08/21

OLS estimates, daily data from 2009/09/15 to 2015/08/21 Dependent Variable: l_BID_ASK_10Y Heteroscedasticity-consistent standard errors

	Coefficient	Std.er	ror t-statistic		p-value	
Const	0.242809	0.164	164	1.4791	0.13933	*
MTS_Volumes	-0.0790045	0.0273	885	-2.8846	0.00397	***
BTP_Volatility	0.0365697	0.0265	292	1.3785	0.16826	
Future_Volumes ²⁵	-0.0271826	0.0104	881	-2.5917	0.00964	***
Spread_IT_DE	Spread_IT_DE 0.02255		3293 2.3409		0.01936	**
BID_ASK_10Y_(1)	0.503941	0.0577	338	8.7287	< 0.00001	***
BID_ASK_10Y_(2)	0.112639	0.0826	049	1.3636	0.17290	
BID_ASK_10Y_(3)	0.0689276	0.0513	519	1.3423	0.17971	
BID_ASK_10Y_(4)	0.214525	0.0459	099	4.6727	< 0.00001	***
R^2	0.83	82038	Adj R ²	2	0	.881428
F(8, 1540)		57.603 P-valu		e(F)	0	.000000
			Durbir	n-Watson	2	.047321

²⁵ The BTP futures traded volume has been first regressed against the MTS traded volume, in order to consider potential multi-collinearity problems. The residuals from the first regression are thus considered in the model specification instead of BTP futures volumes.



(multivariate GARCH-BEKK(1,1) estimates, different frequencies)

Figures show the dynamic correlations behavior between BTP *futures* and related *cheapest-to-deliver* bond estimated through a multivariate GARCH-BEKK(1,1) model. The procedure was iterated across 4 different sampling frequencies at 30, 15, 10 and 5 minutes.

Figure n.3: Correlation curves based on differetent sampling frequencies (*Full sample*)



Red lines represents the median value, the upper and lower boundaries of the box represent values corresponding to the 25th and 75th percentile, while the upper and lower dotted lines extend from minimum to maximum values, excluding the outliers. The outliers are highlighted by asterisks and they represent observations greater than 3 standard deviations.



Figure n.4 : Correlation curves based on differetent sampling frequencies (*Treasury auctions day excluded*)

Red lines represents the median value, the upper and lower boundaries of the box represent values corresponding to the 25th and 75th percentile, while the upper and lower dotted lines extend from minimum to maximum values, excluding the outliers. The outliers are highlighted by asterisks and they represent observations greater than 3 standard deviations.

Figure n.5: Realized Correlations density function

(Gaussian Kernel estimates, different frequencies)



Figures show probability density function estimates adopting a gaussian kernel methodology. Continous line is the correlation between BTP *futures* and the related *cheapest-to-deliver*, dashed lines are density correlations among all the cash securities included in the basket of the deliverable future contract during the period under review.



Figure n.6: Dynamic Volatility estimates between BTP futures and cheapest-to-deliver (Multivariate GARCH estimates, different frequencies)

Figures show the dynamic volatility behavior of the BTP *futures* and the related *cheapest-to-deliver* bond estimated through a multivariate GARCH-BEKK(1,1) model. The procedure was iterated across 4 different sampling frequencies at 30, 15, 10 and 5 minutes. Continuous line is the BTP *futures* volatility, dashed line it's the *cheapest-to-deliver* bond volatility.

Figure n.7 : Realized Volatility density function

(Gaussian Kernel estimates, different frequencies)



Figures show volatility density function estimates adopting a gaussian kernel methodology. Continous line is the BTP *futures* volatility density, dashed lines are density volatilities of all the cash securities included in the basket of the deliverable future contract during the period under review.

Figure n.8: Volatility Spillover

(Granger Causality Test, different frequencies)



Figures show for each sampling frequency, Granger causality test statistics. A boxplot representation is choosen in order to represents F-statistic values obtained across all considered business days (cross-sectional dimension). Values observed above the threshold (99% confidence interval, dashed line) allow to reject the null hypothesis, detecting the presence of a causal relationship (in the sense of Granger) between the two series.

Figure n.9: Volatility *Spillover* on *Bid-Ask*

(Granger causality test, different frequencies)



Figures show for each sampling frequency, Granger causality test statistics. A boxplot representation is choosen in order to represents F-statistic values obtained across all considered business days (cross-sectional dimension). Values observed above the threshold (99% confidence interval, dashed line) allow to reject the null hypothesis, detecting the presence of a causal relationship (in the sense of Granger) between the two series.

- N. 1055 Bank internationalization and firm exports: evidence from matched firm-bank data, by Raffaello Bronzini and Alessio D'Ignazio (February 2016).
- N. 1056 *Retirement, pension eligibility and home production*, by Emanuele Ciani (February 2016).
- N. 1057 The real effects of credit crunch in the Great Recession: evidence from Italian provinces, by Guglielmo Barone, Guido de Blasio and Sauro Mocetti (February 2016).
- N. 1058 *The quantity of corporate credit rationing with matched bank-firm data*, by Lorenzo Burlon, Davide Fantino, Andrea Nobili and Gabriele Sene (February 2016).
- N. 1059 Estimating the money market microstructure with negative and zero interest rates, by Edoardo Rainone and Francesco Vacirca (February 2016).
- N. 1060 Intergenerational mobility in the very long run: Florence 1427-2011, by Guglielmo Barone and Sauro Mocetti (April 2016).
- N. 1061 An evaluation of the policies on repayment of government's trade debt in Italy, by Leandro D'Aurizio and Domenico Depalo (April 2016).
- N. 1062 Market timing and performance attribution in the ECB reserve management framework, by Francesco Potente and Antonio Scalia (April 2016).
- N. 1063 Information contagion in the laboratory, by Marco Cipriani, Antonio Guarino, Giovanni Guazzarotti, Federico Tagliati and Sven Fischer (April 2016).
- N. 1064 EAGLE-FLI. A macroeconomic model of banking and financial interdependence in the euro area, by Nikola Bokan, Andrea Gerali, Sandra Gomes, Pascal Jacquinot and Massimiliano Pisani (April 2016).
- N. 1065 How excessive is banks' maturity transformation?, by Anatoli Segura Velez and Javier Suarez (April 2016).
- N. 1066 *Common faith or parting ways? A time-varying factor analysis*, by Davide Delle Monache, Ivan Petrella and Fabrizio Venditti (June 2016).
- N. 1067 *Productivity effects of eco-innovations using data on eco-patents*, by Giovanni Marin and Francesca Lotti (June 2016).
- N. 1068 *The labor market channel of macroeconomic uncertainty*, by Elisa Guglielminetti (June 2016).
- N. 1069 Individual trust: does quality of public services matter?, by Silvia Camussi and Anna Laura Mancini (June 2016).
- N. 1070 Some reflections on the social welfare bases of the measurement of global income inequality, by Andrea Brandolini and Francesca Carta (July 2016).
- N. 1071 Boulevard of broken dreams. The end of the EU funding (1997: Abruzzi, Italy), by Guglielmo Barone, Francesco David and Guido de Blasio (July 2016).
- N. 1072 Bank quality, judicial efficiency and borrower runs: loan repayment delays in Italy, by Fabio Schiantarelli, Massimiliano Stacchini and Philip Strahan (July 2016).
- N. 1073 Search costs and the severity of adverse selection, by Francesco Palazzo (July 2016).
- N. 1074 Macroeconomic effectiveness of non-standard monetary policy and early exit. A model-based evaluation, by Lorenzo Burlon, Andrea Gerali, Alessandro Notarpietro and Massimiliano Pisani (July 2016).
- N. 1075 *Quantifying the productivity effects of global sourcing*, by Sara Formai and Filippo Vergara Caffarelli (July 2016).
- N. 1076 Intergovernmental transfers and expenditure arrears, by Paolo Chiades, Luciano Greco, Vanni Mengotto, Luigi Moretti and Paola Valbonesi (July 2016).
- N. 1077 A "reverse Robin Hood"? The distributional implications of non-standard monetary policy for Italian households, by Marco Casiraghi, Eugenio Gaiotti, Lisa Rodano and Alessandro Secchi (July 2016).

^(*) Requests for copies should be sent to:

Banca d'Italia – Servizio Studi di struttura economica e finanziaria – Divisione Biblioteca e Archivio storico – Via Nazionale, 91 – 00184 Rome – (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

2014

- G. M. TOMAT, *Revisiting poverty and welfare dominance*, Economia pubblica, v. 44, 2, 125-149, **TD No. 651** (December 2007).
- M. TABOGA, *The riskiness of corporate bonds*, Journal of Money, Credit and Banking, v.46, 4, pp. 693-713, **TD No. 730 (October 2009).**
- G. MICUCCI and P. ROSSI, *Il ruolo delle tecnologie di prestito nella ristrutturazione dei debiti delle imprese in crisi*, in A. Zazzaro (a cura di), Le banche e il credito alle imprese durante la crisi, Bologna, Il Mulino, **TD No. 763 (June 2010).**
- F. D'AMURI, *Gli effetti della legge 133/2008 sulle assenze per malattia nel settore pubblico*, Rivista di politica economica, v. 105, 1, pp. 301-321, **TD No. 787 (January 2011).**
- R. BRONZINI and E. IACHINI, Are incentives for R&D effective? Evidence from a regression discontinuity approach, American Economic Journal : Economic Policy, v. 6, 4, pp. 100-134, **TD No. 791** (February 2011).
- P. ANGELINI, S. NERI and F. PANETTA, *The interaction between capital requirements and monetary policy*, Journal of Money, Credit and Banking, v. 46, 6, pp. 1073-1112, **TD No. 801 (March 2011).**
- M. BRAGA, M. PACCAGNELLA and M. PELLIZZARI, *Evaluating students' evaluations of professors,* Economics of Education Review, v. 41, pp. 71-88, **TD No. 825 (October 2011).**
- M. FRANCESE and R. MARZIA, Is there Room for containing healthcare costs? An analysis of regional spending differentials in Italy, The European Journal of Health Economics, v. 15, 2, pp. 117-132, TD No. 828 (October 2011).
- L. GAMBACORTA and P. E. MISTRULLI, *Bank heterogeneity and interest rate setting: what lessons have we learned since Lehman Brothers?*, Journal of Money, Credit and Banking, v. 46, 4, pp. 753-778, **TD No. 829 (October 2011).**
- M. PERICOLI, *Real term structure and inflation compensation in the euro area*, International Journal of Central Banking, v. 10, 1, pp. 1-42, **TD No. 841 (January 2012).**
- E. GENNARI and G. MESSINA, How sticky are local expenditures in Italy? Assessing the relevance of the flypaper effect through municipal data, International Tax and Public Finance, v. 21, 2, pp. 324-344, TD No. 844 (January 2012).
- V. DI GACINTO, M. GOMELLINI, G. MICUCCI and M. PAGNINI, *Mapping local productivity advantages in Italy: industrial districts, cities or both?*, Journal of Economic Geography, v. 14, pp. 365–394, **TD No. 850** (January 2012).
- A. ACCETTURO, F. MANARESI, S. MOCETTI and E. OLIVIERI, Don't Stand so close to me: the urban impact of immigration, Regional Science and Urban Economics, v. 45, pp. 45-56, TD No. 866 (April 2012).
- M. PORQUEDDU and F. VENDITTI, Do food commodity prices have asymmetric effects on euro area inflation, Studies in Nonlinear Dynamics and Econometrics, v. 18, 4, pp. 419-443, TD No. 878 (September 2012).
- S. FEDERICO, *Industry dynamics and competition from low-wage countries: evidence on Italy*, Oxford Bulletin of Economics and Statistics, v. 76, 3, pp. 389-410, **TD No. 879 (September 2012).**
- F. D'AMURI and G. PERI, *Immigration, jobs and employment protection: evidence from Europe before and during the Great Recession,* Journal of the European Economic Association, v. 12, 2, pp. 432-464, TD No. 886 (October 2012).
- M. TABOGA, *What is a prime bank? A euribor-OIS spread perspective*, International Finance, v. 17, 1, pp. 51-75, **TD No. 895 (January 2013).**
- G. CANNONE and D. FANTINO, *Evaluating the efficacy of european regional funds for R&D*, Rassegna italiana di valutazione, v. 58, pp. 165-196, **TD No. 902 (February 2013).**
- L. GAMBACORTA and F. M. SIGNORETTI, *Should monetary policy lean against the wind? An analysis based on a DSGE model with banking*, Journal of Economic Dynamics and Control, v. 43, pp. 146-74, **TD No. 921 (July 2013).**
- M. BARIGOZZI, CONTI A.M. and M. LUCIANI, Do euro area countries respond asymmetrically to the common monetary policy?, Oxford Bulletin of Economics and Statistics, v. 76, 5, pp. 693-714, TD No. 923 (July 2013).
- U. ALBERTAZZI and M. BOTTERO, *Foreign bank lending: evidence from the global financial crisis,* Journal of International Economics, v. 92, 1, pp. 22-35, **TD No. 926 (July 2013).**

- R. DE BONIS and A. SILVESTRINI, *The Italian financial cycle: 1861-2011*, Cliometrica, v.8, 3, pp. 301-334, **TD No. 936 (October 2013).**
- G. BARONE and S. MOCETTI, *Natural disasters, growth and institutions: a tale of two earthquakes, Journal of Urban Economics, v. 84, pp. 52-66, TD No. 949 (January 2014).*
- D. PIANESELLI and A. ZAGHINI, *The cost of firms' debt financing and the global financial crisis*, Finance Research Letters, v. 11, 2, pp. 74-83, **TD No. 950 (February 2014).**
- J. LI and G. ZINNA, *On bank credit risk: sytemic or bank-specific? Evidence from the US and UK*, Journal of Financial and Quantitative Analysis, v. 49, 5/6, pp. 1403-1442, **TD No. 951 (February 2015).**
- A. ZAGHINI, *Bank bonds: size, systemic relevance and the sovereign,* International Finance, v. 17, 2, pp. 161-183, **TD No. 966 (July 2014).**
- G. SBRANA and A. SILVESTRINI, *Random switching exponential smoothing and inventory forecasting,* International Journal of Production Economics, v. 156, 1, pp. 283-294, **TD No. 971 (October 2014).**
- M. SILVIA, Does issuing equity help R&D activity? Evidence from unlisted Italian high-tech manufacturing firms, Economics of Innovation and New Technology, v. 23, 8, pp. 825-854, TD No. 978 (October 2014).

2015

- G. DE BLASIO, D. FANTINO and G. PELLEGRINI, Evaluating the impact of innovation incentives: evidence from an unexpected shortage of funds, Industrial and Corporate Change, v. 24, 6, pp. 1285-1314, TD No. 792 (February 2011).
- M. BUGAMELLI, S. FABIANI and E. SETTE, The age of the dragon: the effect of imports from China on firmlevel prices, Journal of Money, Credit and Banking, v. 47, 6, pp. 1091-1118, TD No. 737 (January 2010).
- R. BRONZINI, The effects of extensive and intensive margins of FDI on domestic employment: microeconomic evidence from Italy, B.E. Journal of Economic Analysis & Policy, v. 15, 4, pp. 2079-2109, TD No. 769 (July 2010).
- A. DI CESARE, A. P. STORK and C. DE VRIES, *Risk measures for autocorrelated hedge fund returns*, Journal of Financial Econometrics, v. 13, 4, pp. 868-895, **TD No. 831 (October 2011).**
- G. BULLIGAN, M. MARCELLINO and F. VENDITTI, *Forecasting economic activity with targeted predictors,* International Journal of Forecasting, v. 31, 1, pp. 188-206, **TD No. 847 (February 2012).**
- A. CIARLONE, *House price cycles in emerging economies*, Studies in Economics and Finance, v. 32, 1, **TD No. 863 (May 2012).**
- D. FANTINO, A. MORI and D. SCALISE, Collaboration between firms and universities in Italy: the role of a firm's proximity to top-rated departments, Rivista Italiana degli economisti, v. 1, 2, pp. 219-251, TD No. 884 (October 2012).
- A. BARDOZZETTI and D. DOTTORI, *Collective Action Clauses: how do they Affect Sovereign Bond Yields?*, Journal of International Economics, v 92, 2, pp. 286-303, **TD No. 897 (January 2013).**
- D. DEPALO, R. GIORDANO and E. PAPAPETROU, *Public-private wage differentials in euro area countries:* evidence from quantile decomposition analysis, Empirical Economics, v. 49, 3, pp. 985-1115, **TD No. 907 (April 2013).**
- G. BARONE and G. NARCISO, Organized crime and business subsidies: Where does the money go?, Journal of Urban Economics, v. 86, pp. 98-110, **TD No. 916 (June 2013).**
- P. ALESSANDRI and B. NELSON, *Simple banking: profitability and the yield curve,* Journal of Money, Credit and Banking, v. 47, 1, pp. 143-175, **TD No. 945 (January 2014).**
- M. TANELI and B. OHL, *Information acquisition and learning from prices over the business cycle*, Journal of Economic Theory, 158 B, pp. 585–633, **TD No. 946 (January 2014).**
- R. AABERGE and A. BRANDOLINI, *Multidimensional poverty and inequality*, in A. B. Atkinson and F. Bourguignon (eds.), Handbook of Income Distribution, Volume 2A, Amsterdam, Elsevier, TD No. 976 (October 2014).
- V. CUCINIELLO and F. M. SIGNORETTI, *Large banks,loan rate markup and monetary policy*, International Journal of Central Banking, v. 11, 3, pp. 141-177, **TD No. 987 (November 2014).**
- M. FRATZSCHER, D. RIMEC, L. SARNOB and G. ZINNA, *The scapegoat theory of exchange rates: the first tests*, Journal of Monetary Economics, v. 70, 1, pp. 1-21, **TD No. 991 (November 2014).**

- A. NOTARPIETRO and S. SIVIERO, Optimal monetary policy rules and house prices: the role of financial frictions, Journal of Money, Credit and Banking, v. 47, S1, pp. 383-410, TD No. 993 (November 2014).
- R. ANTONIETTI, R. BRONZINI and G. CAINELLI, *Inward greenfield FDI and innovation*, Economia e Politica Industriale, v. 42, 1, pp. 93-116, **TD No. 1006 (March 2015).**
- T. CESARONI, *Procyclicality of credit rating systems: how to manage it*, Journal of Economics and Business, v. 82. pp. 62-83, **TD No. 1034 (October 2015).**
- M. RIGGI and F. VENDITTI, *The time varying effect of oil price shocks on euro-area exports,* Journal of Economic Dynamics and Control, v. 59, pp. 75-94, **TD No. 1035 (October 2015).**

2016

- E. BONACCORSI DI PATTI and E. SETTE, Did the securitization market freeze affect bank lending during the financial crisis? Evidence from a credit register, Journal of Financial Intermediation, v. 25, 1, pp. 54-76, TD No. 848 (February 2012).
- M. MARCELLINO, M. PORQUEDDU and F. VENDITTI, Short-Term GDP Forecasting with a mixed frequency dynamic factor model with stochastic volatility, Journal of Business & Economic Statistics, v. 34, 1, pp. 118-127, TD No. 896 (January 2013).
- M. ANDINI and G. DE BLASIO, *Local development that money cannot buy: Italy's Contratti di Programma,* Journal of Economic Geography, v. 16, 2, pp. 365-393, **TD No. 915 (June 2013).**
- L. ESPOSITO, A. NOBILI and T. ROPELE, *The Management of Interest Rate Risk During the Crisis: Evidence from Italian Banks*, Journal of Banking & Finance, v. 59, pp. 486-504, **TD No. 933 (September 2013).**
- F. BUSETTI and M. CAIVANO, The Trend–Cycle Decomposition of Output and the Phillips Curve: Bayesian Estimates for Italy and the Euro Area, Empirical Economics, V. 50, 4, pp. 1565-1587, TD No. 941 (November 2013).
- M. CAIVANO and A. HARVEY, *Time-series models with an EGB2 conditional distribution*, Journal of Time Series Analysis, v. 35, 6, pp. 558-571, **TD No. 947 (January 2014).**
- G. ALBANESE, G. DE BLASIO and P. SESTITO, *My parents taught me. evidence on the family transmission of values,* Journal of Population Economics, v. 29, 2, pp. 571-592, **TD No. 955 (March 2014).**
- R. BRONZINI and P. PISELLI, *The impact of R&D subsidies on firm innovation*, Research Policy, v. 45, 2, pp. 442-457, **TD No. 960 (April 2014).**
- L. BURLON and M. VILALTA-BUFI, A new look at technical progress and early retirement, IZA Journal of Labor Policy, v. 5, **TD No. 963 (June 2014).**
- A. BRANDOLINI and E. VIVIANO, *Behind and beyond the (headcount) employment rate,* Journal of the Royal Statistical Society: Series A, v. 179, 3, pp. 657-681, **TD No. 965 (July 2015).**
- D. DOTTORI and M. MANNA, *Strategy and Tactics in Public Debt Management*, Journal of Policy Modeling, v. 38, 1, pp. 1-25, **TD No. 1005 (March 2015).**
- A. CALZA and A. ZAGHINI, *Shoe-leather costs in the euro area and the foreign demand for euro banknotes,* International Journal of Central Banking, v. 12, 1, pp. 231-246, **TD No. 1039 (December 2015).**
- E. CIANI, *Retirement, Pension Eligibility and Home Production,* Labour Economics, v. 38, pp. 106-120, **TD** No. 1056 (March 2016).
- L. D'AURIZIO and D. DEPALO, An Evaluation of the Policies on Repayment of Government's Trade Debt in *Italy*, Italian Economic Journal, v. 2, 2, pp. 167-196, **TD No. 1061 (April 2016).**

FORTHCOMING

- S. MOCETTI, M. PAGNINI and E. SETTE, *Information technology and banking organization*, Journal of Financial Services Research, **TD No. 752 (March 2010).**
- F BRIPI, *The role of regulation on entry: evidence from the Italian provinces*, World Bank Economic Review, **TD No. 932 (September 2013).**

- G. DE BLASIO and S. POY, *The impact of local minimum wages on employment: evidence from Italy in the* 1950s, Regional Science and Urban Economics, **TD No. 953 (March 2014).**
- A. L. MANCINI, C. MONFARDINI and S. PASQUA, *Is a good example the best sermon? Children's imitation of parental reading*, Review of Economics of the Household, **TD No. 958 (April 2014).**
- L. BURLON, *Public expenditure distribution, voting, and growth,* Journal of Public Economic Theory, **TD** No. 961 (April 2014).
- G. ZINNA, Price pressures on UK real rates: an empirical investigation, Review of Finance, TD No. 968 (July 2014).
- A. BORIN and M. MANCINI, Foreign direct investment and firm performance: an empirical analysis of *Italian firms*, Review of World Economics, **TD No. 1011 (June 2015).**
- F. CORNELI and E. TARANTINO, *Sovereign debt and reserves with liquidity and productivity crises*, Journal of International Money and Finance, **TD No. 1012 (June 2015).**